Author's response to reviews

Title: Serum hyaluronate as a non-invasive marker of hepatic fibrosis and inflammation in HBeAg-negative chronic hepatitis B

Authors:

Ghodrat Montazeri MD (montazer@ams.ac.ir)
Arezoo Estakhri MD (Arezooestakhri@yahoo.com)
Mehdi Mohamadnejad MD (mehdim@ams.ac.ir)
Negin Nouri MD (neginoo@ams.ac.ir)
Farhad Montazeri MD (montazer@ams.ac.ir)
Ashraf Mohamadkhani (Ashraf@ams.ac.ir)
Mohammad Hossein Derakhshan MD (mhd@ams.ac.ir)
Farhad Zamani MD (zamanif@hotmail.com)
Shahram Samiee (shsamie@ibto.ir)
Reza Malekzadeh MD (malek@ams.ac.ir)

Version: 6 Date: 8 October 2005

Author's response to reviews:

Dear Editors
Thank you very much for the acceptance of our manuscript. The only change made in the manuscript was the correction of the name of the 4th author.
Incorrect: Negin Noori; Correct: Negin Nouri
Also, the reference 8 was missed in the second revised manuscript which was added to this final version.

Best Regards
Prof. Malekzadeh